Daily Status Wednesday 10-31-12

Weather

Clear skies and light winds. Temperatures are 5C.

Status

Qsoilc went down again. Steve O. believes it is a temperature related problem. Vdsmé still down.

Tom is coming up today so we should get the 10m krypton repaired. It appears the 10m sonic on the

main tower knew we were going to pay a visit because it has been acting up throughout the night. The

daig has favored a value of 15 verses 0. Since this based on the background it is hard to say if it was continuous through
the night or just a short occurrence. According to Gordon's notes this means a skipped sampled.

Changed the pocketec this morning.

Update.

At about 7:45am the 10m sonic on main started behaving. Maybe some spider was building a web.

Will investigate when we replace the krpyton.

Here is an update from Gordon on the 10m sonic activity: There are a bunch of 4's and 8's around 00:24 UTC (18:24 MDT). Then it
was OK, until 13:14 UTC (07:14) when it started having lots of problems.

(bird poop?)

Afternoon: Site C had gone down. Gordon was able to ping the etherant but no further.

17:30:

kh20.10m ok, station C20 ok.

Vdsm is RIP for stations 1, 2, 6..

Sonic at 17.1m has been intermittently setting diagabits = 4.

Qsoil.c is okay at the moment.
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